Author’s response to reviews

Title: Validating a transnational fracture treatment registry using a standardized method

Authors:

Jasper Frese (jasper.frese@yahoo.de)
Annalice Gode (annalice.gode@uksh.de)
Gerhard Heinrichs (gerhard.heinrichs@uksh.de)
Armin Will (armin.will@uksh.de)
Arndt-Peter Schulz (schulz@biomechatronics.de)

Version: 2 Date: 28 Oct 2019

Author’s response to reviews:

@Mrs. Ojeda- Thies
The conclusion section is now significantly shorter and most of it has been moved to the discussion. Keywords for the initial search are mentioned in line 83.

@Mr. Stausberg
The intended platform of Nonnemacher et al.’s publication is mentioned in line 97. Confidence intervals are now reported at the end of the result’s section. The semantic difference of the method having been “published” and “applied” has been clarified in line 336. Hyphens in „BFCC- project“ are now avoided according to the translation service consulted.